Author's response to reviews

Title: The foot-health of people with diabetes in a regional Australian population: a prospective clinical audit

Authors:

Byron M Perrin (b.perrin@latrobe.edu.au)
Marcus J Gardner (mgardner@bendigohealth.org.au)
Susan R Kennett (susankennett@bchs.com.au)

Version: 4 Date: 21 February 2012

Author's response to reviews: see over
Editor’s comments:

1. Review the manuscript thoroughly and ensure consistency in spacing between symbols (e.g. \( p < 0.001 \) versus \( p<0.001 \)).

Response: There were several instances of this issue. These have been checked and corrected.

2. Review Tables 4, 5 and 6: ensure consistency in use of decimal points (e.g. Table 4, 100 should be 100.0)

Response: This example has been corrected and the text checked for similar examples.

3. Review Table 5 and add space between ‘to’ and ‘23.3’.

Response: Completed.